Author's response to reviews

Title: Internet use, needs, and expectations of web-based information and communication in childbearing women with type 1 diabetes

Authors:

Carina Sparud-Lundin (carina.s-lundin@fhs.gu.se)
Agneta Ranerup (ranerup@ituniv.se)
Marie Berg (marie.berg@gu.se)

Version: 6 Date: 1 July 2011

Author's response to reviews: see over
Gothenburg 1th July, 2011

Dear Editor,

We are delighted to the acceptance of our paper and have made all revisions suggested in the acceptance letter. Please let us know if there is anything missing or misunderstood.

On behalf of the authors,

Carina Sparud Lundin